Author’s response to reviews

Title: Increasing smoking cessation care across a network of hospitals: an implementation study

Authors:

Carolyn Slattery (carolyn.slattery@hnehealth.nsw.gov.au)
Megan Freund (megan.freund@hnehealth.nsw.gov.au)
Karen Gillham (karen.gillham@hnehealth.nsw.gov.au)
Jenny Knight (jenny.knight@hnehealth.nsw.gov.au)
Luke Wolfenden (luke.wolfenden@hnehealth.nsw.gov.au)
Alessandra Bisquera (alessandra.bisquera@hmri.com.au)
John Wiggers (john.wiggers@hnehealth.nsw.gov.au)

Version: 2
Date: 25 March 2015

Author’s response to reviews:

As requested:
- a competing interests section has been included at the end of the manuscript, before the reference list.
- an acknowledgement section has been included at the end of the manuscript before the reference list.
- an Authors’ Contributions section has been included at the end of the manuscript, before the reference list.
- the figures have been uploaded as separate files via the online submission system.
- a figure title and legend section has been included after the reference list.